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D5 Data Taking

Delivered  Recorded  Efficiency Comment
Lum (pb")  Lum (pb™) (%)
Used downtime to de-rime heat exchanger. 30
27 Oct 2.5¢ 2.13 84 minutes down after spike in losses tripped silicon.
28 Oct 2.45 2.27 93 40 minutes for silicon rack trip
29 Oct 333 310 03 Thanks to Steve Cha.pp.a for making our
standard clock timing change.
30 Oct 8.14 7.70 95
31 Oct 9.10 8.69 95 Record initial luminosity @ DO, 315.06E30 (#3)
75 minutes down for problem with L2 and muon
1 Nov 7.29 6.40 38 system.
Access for minor jobs.
2Nov 847 6.13 » Record initial luminosity @ DO, 316.80E30 (42)
27 Oct -
> Nov 39.3 36.4 93
October 148 135 92
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DD Data Taking

Delivered  Recorded  Efficiency Comment
Lum (pb?)  Lum (pb?) (%)
959 87 Cryo problems cause our solenoid and fiber tracker to
3 Nov 11.04 6.60 60 be down for ~7 hours and affect our luminosity
(6.60) (60) measurement. Record daily delivered Luminosity.
4 Nov 9.52 9.06 95 Record initial luminosity @ DO, 330.15E30
5 Nov 3.92 3.72 95 Access to recover from power outage
6 Nov 0.00 0.00 Run cosmics to exercise system when possible
/ Nov 0.00 0.00 Access to recover from power outage
8 Nov 0.00 0.00 Access to recover CAL channels
9 Nov 0.00 0.00
22.4 91 .
3-9 Nov 24.5 (19.4) 79) (Data with full detector)

Congratulations to AD for Luminosity records.
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Cryo Problems

* Last Monday at 17:08 a helium compressor, mycom #2, tripped. Mycom #1
would not start and the DO solenoid quenched.

— Mycom #2 was restarted after 15 minutes.

= 19:50, the mycom tripped again and could not be restarted. DO was
hooked up to the Tev headers.

* Tuesday, mycom #1 was repaired.
* Wednesday, during the power outage, the DO solenoid quenched.

— DO disconnected from the Tev header and went back to mycom #1.

® Currently we are still running on mycom #1 with a backup of the Tev
headers. Mycom #2 is still down.

— Thanks to Cryo support for all their help in keeping the DO solenoid

cold.
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Power outage

* From the power outages, DO lost

— a preamp power supply,
— one AFE board,
— one computer,

— a disk array power supply.

* So we fared better than in the scheduled outages during
the October shutdown.
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Daily Data Taking Efficiency
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